ss9-Jun-03
Week 24: (9/06 to 14/06)

SM18 Operation SCHEDULE (Monday Morning) Meeting 

· Test Coordinator this week:   O.Berrig

· MTA Technician on call:  (16-3602) F.Chiusano.

1. Planning for this week : 
· TBA1:   1022  :  Disconnection and 3014 is coming ( 1 run foreseen).
· TBA2:   1016 Test at 1.9 K (one run foreseen)
· TBC2 :  1026 :  Cool down and Test at 1.9 K (1 run foreseen)
·  TBC1 : 1024 test finished.
2.  Problems from Operation Last week ( Marco+ Ewald)
· Quench at 1.5 kA:  problem with pre trigger : setting modified by Giorgio reducing the sampling frequency.

· IAP at warm : Tuning performed by Giorgio (T gauges not correct values, V tapes errors., gain offsets..).

· AQA: Jumps in the QH discharges,

· AQA : faulty quench on 1022 during DRQ (no current read).

· Excel tables : problems to save the modifications related to the worksheets.

· TEMA on C bench : Swap of the names (C1/C2) of the magnet for a load line

· TBA1: Correctors measurements.  Problem in powering. Setting MM not properly  prepared in AP2 . Dynamic gain table selected ant it does not exist so G=0!.

· MM problems : Shaft 23 and 24 (3013) 2 sectors with signs inverted for odds multipoles. 

Shaft 25 : 3 sectors wrong for b1 (aperture 1).

3. Situation From Cryogenics:  J-P Lamboy
4. Other General Issues.

